Author's response to reviews

Title: Development and promotion of a national website to improve dissemination of information related to the prevention of mother-to-child HIV transmission (PMTCT) in Tanzania

Authors:

  Gudila Stephan (gudilastephan@yahoo.com)
  Mary Jo Hoyt (hoyt@sn.rutgers.edu)
  Deborah Storm (stormds@sn.rutgers.edu)
  Sylvia Shirima (shrimsy@fxbtanzania.org)
  Charles Matiko (Matiko@fxbtanzania.org)
  Emmanuel Matechi (matechi@fxbtanzania.org)

Version: 6 Date: 22 September 2015

Author's response to reviews: see over
Author's response to reviews

Title: Development and promotion of a national website to improve dissemination of information related to the prevention of mother-to-child HIV transmission (PMTCT) in Tanzania

Authors:
Gudila Stephan (gudilastephan@yahoo.com)
Mary Jo Hoyt (hoyt@sn.rutgers.edu)
Deborah S. Storm (stormds@sn.rutgers.edu)
Sylvia Shirima (sshirima@fxbthealth.org)
Charles Matiko (Matiko@fxbthealth.org)
Emmanuel Matechi (matechi@fxbthealth.org)

Version 5.0 Date: September 21, 2015

Author's response to the message from the editorial office:
- Do you have explicit permission to reproduce and publish figure 1, which contains identifiable images of individuals, including a minor. If you do, explicitly state so in the figure legend. If you do not you will need to obtain permission or the images must be removed.

We purchased the image with the following Rights Granted defined on the paid invoice as follows: RIGHTS GRANTED: For EDITORIAL (NON-COMMERICALS) USE ONLY. 100,000 maximum prints or impressions (including collateral and multimedia uses). Unlimited use on one Website permitted. Worldwide distribution permitted.

- Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals). It is important that your files are correctly formatted.

We have followed formatting instructions per the link and template provided. As per template:
- Title page, page numbers and Figure and Table headers altered to conform with template
- Major headers are now Arial 16 pt font; secondary headers Arial 11 pt font.
- Switched to A4 from standard U.S letter size.
- All text (other than headers) now Times New Roman 12 point font